\documentclass{JAC2003}

%%
%%  This file was updated in March 2011 by T. Satogata to be in line with Word templates.
%%
%%  Use \documentclass[boxit]{JAC2003}
%%  to draw a frame with the correct margins on the output.
%%
%%  Use \documentclass[acus]{JAC2003}
%%  for US letter paper layout
%%

\usepackage{graphicx}
\usepackage{booktabs}

%%
%%   VARIABLE HEIGHT FOR THE TITLE BOX (default 35mm)
%%

\setlength{\titleblockheight}{55mm}

\begin{document}
\begin{thebibliography}{9}   % Use for  1-9  references
%\begin{thebibliography}{99} % Use for 10-99 references

\bibitem{ler-spring8-1}
H. Tanaka et. al., Proc. of IPAC2016, (2016) 2867-2870, Busan, Korea.

\bibitem{ler-spring8-2}
T. Watanabe et. al., Proc. of IPAC2016, (2016) 1093-1095, Busan, Korea.

\bibitem{ler-nsls2}
NSLS-II preliminary design report, http://www.bnl.gov/nsls2/project/PDR/.

\bibitem{ler-max4}
The MAX IV Detailed Design Report, http://www.maxlab.lu.se/maxlab/max4/DDR\_public/index.html.

\bibitem{ler-sirius}
L. Liu et. al., Proc. of IPAC2013, (2013) 1874-1876, Shanghai, China.

\bibitem{vwm-alignment}
A. Jain et. al., Proc, of IWAA2008, (2008) 1-6, Tsukuba, Japan.

\bibitem{vwm-theory}
A. Temnykh et. al., Nucl. Inst. and Meth. A399 (1997) 185-194.

\bibitem{vwm-field-meas}
S. Kashiwagi et. al., Proc. of IPAC2012, (2012) 732-734, New Orleans, Louisiana, USA.

\bibitem{vwm-feedback}
K. Fukami et. al., Proc. of IPAC2014, (2014) 277-279, Dresden, Germany.

\bibitem{sag-meas}
S. Matsui et. al., Proc. of 13th Annual Meeting of Particle Accelerator Society of Japan,
 (2016) 277-279, Chiba, Japan (in Japanese).
\end{thebibliography}

\title{ESTIMATION OF MAGNET ALIGNMENT ACCURACY \\FOR SPRING-8 UPGRADE
 USING RESONANCE-FREQUENCY TRACKED VIBRATING WIRE}

\author{K. Fukami$^{1),2)}$\thanks{fukami@spring8.or.jp}, N. Azumi$^{1),2)}$, T. Kai$^{3)}$, H. Kimura$^{1),2)}$,
J. Kiuchi$^{3)}$, S. Matsui$^{2)}$, \\T. Watanabe$^{1),2)}$, and C. Zhang$^{1)}$\\
1) Japan Synchrotron Radiation Research Institute, Hyogo, 679-5198, JAPAN\\
2) RIKEN SPring-8 Center, Hyogo, 679-5148, JAPAN\\
3) SPring-8 Service Co. Ltd., Hyogo, 679-5165, JAPAN}

\maketitle

\begin{abstract}

SPring-8 major upgrade, SPring-8-II, is being designed.
The new storage ring requires alignment errors of 25 [$\mu$m] (1$\sigma$) or less for multi-pole magnets
 on a straight line between adjacent two bending magnets~\cite{ler-spring8-1}.
Using a test bench consisted of five multi-pole magnets with typical field gradients, we have demonstrated
 an alignment of these magnets by introducing a vibrating wire method (VWM) without conventional fiducialization.
A resolution was better than 1 [$\mu$m] for a measurement of a magnetic center, however, systematic errors
 caused by a wire-sag, and caused by background fields, were negligibly small.
The background fields were canceled by counter dipole and quadrupole magnets.
In order to observe a drift of the magnetic center caused by a temperature rise of a water-cooling magnet,
 and caused by a deformation after installation, etc., we developed a resonance-frequency tracked vibrating wire
 for a sustainable measurement.
Actual accuracy of our alignment procedure, including the drift, will be described in this paper.
\end{abstract}

\section{Introduction}
Low emittance storage rings are being constructed at synchrotron radiation facilities in the world~\cite{ler-spring8-1}--\cite{ler-sirius}.
Because an alignment tolerance is one of most important issues, high alignment precision of micro-meter order
 enhances flexibility in the design of the low emittance ring.
For the SPring-8-II, multipole magnets between adjacent two bending magnets will be installed on a common girder,
 and will be aligned out of the machine tunnel.
After the alignment, the magnets fixed on the common girder will be transported to the machine tunnel, and will be installed.
Therefore, it is necessary to suppress a total error, including a change and a drift of the magnetic center
 caused by a transportation and a deformation. 
 
A vibrating wire method (VWM) has been used for such a high precision alignment~\cite{vwm-alignment}.
A tensioned wire excited with AC current is placed along a longitudinal direction.
A field profile of a multi-pole magnet in the vicinity of its magnetic center can be measured 
 by detecting a vibration of the wire at its resonance frequency.
 
The resonance frequency easily drifts due to a change in an ambient air temperature.
To trace the resonance frequency at all times, we developed a frequency feedback system.
It is necessary to trace the resonance even if the wire is placed in the vicinity of the magnetic center,
 where the magnetic field is nearly zero.
Here, we propose to install one additional tensioned wire parallel to the original wire.
Using the system, we aim to establish an {\it in-situ} alignment procedure with actual accuracy of micro-meter order.

\section{Principle}
When a tensioned wire is supplied with AC current in a magnetic field, vibrations of the wire in horizontal and vertical directions
are generated by a vertical flux component {\it B$_{y}$}, and a horizontal flux component {\it B$_{x}$}, respectively. 
An amplitude {\it A$_{n}$($\omega$)}, and a phase {\it $\phi$$_{n}$($\omega$)} of the vibration near its resonance frequency
 are shown as~\cite{vwm-theory},  

\begin{equation}\label{amplitude}
A_{n}(\omega)=\frac{a_{n}}{\sqrt{(\omega^{2}-b_{n}^{2})^{2}+c_{n}^{2}\omega^{2}}},
\end{equation}
\begin{equation}\label{phase}
\phi_{n}(\omega)=\tan^{-1}{\frac{c_{n}\omega}{\omega^{2}-b_{n}^{2}}},
\end{equation} 
where {\it n} is mode number, {\it b$_{n}$}, and {\it c$_{n}$} are resonance frequency [rad/s] and damping constant [s$^{-1}$], respectively.
Magnetic field at the wire is estimated by coefficient {\it a$_{n}$} [m/s$^{2}$]~\cite{vwm-field-meas}. 
The frequency of the AC current is always fixed to the resonance frequency ({\it $\omega$=b$_{n}$}), the field is estimated by only
 one amplitude {\it A$_{n}$(=a$_{n}$/b$_{n}$c$_{n}$)} without a fitting by Eq.~\ref{amplitude}.

\section{Frequency tracking}
Change in the resonance frequency can be estimated by measuring the phase of the wire vibration
 because the phase is $\pi$/2 [rad] at the resonance~\cite{vwm-feedback}.
From Eq.~\ref{phase}, the change in the resonance frequency $\Delta\omega$ [rad/s] is expressed as following equation. 

\begin{equation}\label{frequency_change}
\Delta\omega\cong\frac{c}{2\tan\phi(\omega)}.
\end{equation} 
Set frequency of the supplied current was traced to the resonance frequency by controlling the phase of itself using Eq.~\ref{frequency_change}.
We shall call this process ``basic feedback''.

When the wire is set in the vicinity of the magnetic center, the signal strength is not enough for the basic feedback.
In addition, a polarity of the phase is frequently changed in the position.
Second wire was installed in parallel to the above the signal wire as a designated wire for the feedback,
 which we shall call ''feedback wire``.
Distances between the signal wire and the feedback wire were 5 [mm] in horizontal and vertical directions.        
The set frequency of the supplied current of the signal wire was traced to the resonance frequency by controlling
 the phase of the feedback wire.
We shall call this process ''advanced feedback``.

\section{Experiment}
\subsection{Sag Correction and Kink Survey}
Vertical position of a tensioned wire is shown a catenary curve in longitudinal direction.
A sag, {\it S(s)}, which is defined as a vertical displacement from the both end of the wire, becomes maximum at the center of the wire.
The maximum sag {\it S$_{max}$} is expressed by the fundamental resonance frequency {\it $\omega$$_{1}$} as~\cite{vwm-theory},

\begin{equation}\label{sag}
S_{max}=\frac{g}{32(\omega_{1}/2\pi)^{2}},
\end{equation}
where, {\it g} [m/s$^{2}$] is the acceleration due to gravity.
Therefore, the sag can be corrected by measuring the resonance frequency.

In order to confirm, vertical position of a wire was measured by a Hydrostatic Leveling System (FOGALE, HLS),
 and a Wire Positioning System (FOGALE, WPS) on a precise stone surface plate~\cite{sag-meas}.
Wire length was 4.433 [m].
One end of the wire was fixed on the surface plate.
Two kinds of tensions, 1.75 [kgw], and 2.00 [kgw], were applied.
Nominal maximum sag were 0.36 [mm], and 0.32 [mm], respectively.
Two sensors of the HLS were placed on the surface plate to measure a figure error of the surface plate.
One sensor was fixed as a reference point at the center of the surface plate in longitudinal direction.
The other sensor was placed just below the tensioned wire.
To measure a relative height of the wire, a sensor of the WPS was put on the top of the other sensor of the HLS.
A moveable unit of the sensors are shown in Fig.~\ref{sag_photo}.
The unit was moved statically along the wire,
 and then, relative heights of the wire were measured at seven points in longitudinal direction.      

A s-axis was defined along the wire in longitudinal direction, and the center of the wire was defined as {\it s}=0.
A height reference line, {\it y}=0, was defined as a line passing through two wire positions at  {\it s}=$\pm$1.8 [m].
A difference of absolute height between the two points is not important.
Even if the difference is equal to 5 [mm], a change in the theoretical sag due to the difference are less than 2.1$\times$10$^{-4}$ [$\mu$m].
In case of the test, the absolute height at the two points were roughly corresponded within 1 [mm]. 

\begin{figure}[htbp]
\begin{center}
\includegraphics*[width=80mm]{TUO_fukami_sag_photo.eps}
\end{center}
\vspace*{-8mm}
\caption{Moveable unit of sensors of HLS and WPS placed on a precise stone surface plate.
Two sensors of the HLS were prepared.
One more sensor of the HLS was fixed as a reference point (out of the figure).}
\label{sag_photo}
\end{figure}

\subsection{Measurement of Magnetic Center}
Three quadrupole magnets and two sextupole magnets were prepared as test magnets.
Outline of the test magnets and a common girder are shown in Fig.~\ref{outline}.
A 4.871-[m] long Be-Cu wire with 0.2-[mm] diameter was prepared as a field signal wire.
One end of the wire was fixed through a pulley.
A load of 2.00 [kg] was hung at the other end of the wire through a pulley.
A ball-bearing pulley was adopted at the loaded end to keep the tension.  
Nominal fundamental resonance frequency is 2$\pi$$\times$28.3 [rad/s].

Wire girders at the both ends was made of a super-invar material to suppress a change in the position
 due to a fluctuation of the girder temperature.
At the both end of the wire girder, the pulleys were placed on x-y stages with a resolution
 of 0.1 [$\mu$m] to scan the wire in a transverse plane.
The wire was moved in parallel using these stages.
In order from the loaded end, the test magnets are called {\it Q$_{1}$, S$_{2}$, Q$_{3}$, S$_{4}$, and Q$_{5}$}.
Symbols {\it ``Q''} and {\it ``S''} indicate quadrupole and sextupole magnets, respectively.
Specifications of these magnets are listed in Table ~\ref{test_magnet_spec}.
In this table, {\it l$_{p}$} is the pole length and {\it l$_{e}$} is the effective length, which is defined Eq.~\ref{effective_length}.

\begin{figure*}[tb]
    \centering
    \includegraphics*[width=140mm]{TUO_fukami_outline_vwm_system.eps}
    \caption{Outline of the test magnets and test common girder.
    The wire is fixed through a pulley at one end.
    A load is hung through a pulley at the other end.
    These pulleys are placed on x-y stages to scan the wire.
    The ``signal wire'' and the ``feedback wire'' were tensioned in parallel with a distance of 5 [mm] in {\it x} and {\it y} directions.
    The {\it x} and {\it s} axes point to the north and the east.
    Wire length is 4.871 [m]. Tension is 2.00 [kgw]. Maximum sag is 0.383 [mm]. Nominal fundamental resonance frequency is 28.3 [Hz]
    Bottom of figure is shown block diagram of vibration amplitude measurement.}
    \label{outline}
\end{figure*}

\begin{table}[hbt]
\begin{center}
\caption{Specification of test magnets}
\begin{tabular}{|c|c|c|c|}
\hline
\textbf{Name} & \textbf{Bore dia.} & \textbf{Max. gradient} & \textbf{Length}\\
\textbf{} & \textbf{[mm]} & \textbf{[T/m] or [T/m$^{2}$]} & \textbf{{\it l$_{p}$, l$_{e}$} [mm]}\\ \hline
Q$_{1}$ & 85 & 17.0 @536[A] & 310, 138\\
Q$_{3}$ & 85 & 17.6 @552[A] & 970, 954\\
Q$_{5}$ & 85 & 17.4 @544[A] & 470, 222\\
S$_{2}$ & 92 & 420 @300[A] & 270, 211\\
S$_{4}$ & 92 & 420 @300[A] & 270, 224\\ \hline
\end{tabular}
\label{test_magnet_spec}
\end{center}
\end{table}

Supplied current of the wire was generated by an arbitrary waveform generator.
An amplitude of the current was 98 [mA$_{rms}$]. 
Wire positions in horizontal and vertical directions were measured by laser sensors (KEYENCE Ltd, LS-9000D) placed
 at the longitudinal position of 1.83 [m] from the fixed end.
The amplitude and the phase of the vibration at the set frequency were picked up by a lock-in-amplifier.
To cancel a background field, e.g., geomagnetism, a pair of counter-dipole magnets ({\it Sth}, and {\it Stv}) were installed.
Maximum integrated field of these magnet are 1.5 [mTm].

Using only a fundamental resonance frequency, flux densities in the test magnets were estimated
 by measuring the amplitude at the resonance frequency.  
Magnetic centers of quadrupole and sextupole magnets were measured by scanning the wire position in horizontal
 and vertical directions.
Definition of coordinate axes are also shown in Fig.~\ref{outline}.
Only a target magnet for the measurement was excited. 

\subsubsection{Quadrupole Magnet}
The magnetic center of a quadrupole magnet in transverse direction ({\it x$_{0}$}, {\it y$_{0}$}) is defined  by the point,
 where the horizontal dipole and vertical dipole are zero.    
Horizontal and vertical distributions of integrated flux densities {\it B$_{x}$l$_{e}$} and {\it B$_{y}$l$_{e}$} [Tm] in the magnet
 are expressed as,

\begin{equation}\label{bx_qm}
B_{x}(y)l_{e}=G_{q}(y-y_{0})l_{e},
\end{equation}
\begin{equation}\label{by_qm}
B_{y}(x)l_{e}=G_{q}(x-x_{0})l_{e},
\end{equation}
where, {\it G$_{q}$} [T/m] is a gradient of a quadrupole magnet.
 The length {\it l$_{e}$} is an effective length considered a sensitivity of the vibration to the field, which is defined as,

\begin{equation}\label{effective_length}
l_{e}\equiv\int_{s_u}^{s_d}sin(\frac{n\pi s}{L})ds,
\end{equation}
where, {\it L} [m] is the wire length, {\it s$_{u}$} and {\it s$_{d}$} [m] are longitudinal positions of upstream and downstream edges
 of the magnet, respectively.
The effective lengths for the fundamental resonance are also listed in Table ~\ref{test_magnet_spec}.
Background field components of {\it $\delta$b$_{x}$L} and {\it $\delta$b$_{y}$L} [Tm] are added to Eq.~\ref{bx_qm} and Eq.~\ref{by_qm}.
These components were canceled by {\it Sth} and {\it Stv}.

The vibration amplitude {\it A$_{x}$} and {\it A$_{y}$} are in proportional to integrated flux densities,
 {\it B$_{y}$l$_{e}$} and {\it B$_{x}$l$_{e}$}, respectively.
By scanning the wire in horizontal and vertical directions, the coordinates of the magnetic center were found out.
The amplitudes, {\it A$_{x}$} and {\it A$_{y}$}, at the magnetic center are equal to zero.
A scan step was 2 [$\mu$m] in the range of $\pm$10 [$\mu$m] in both directions.
Excitation current of all the target magnets was 300 [A].
Nominal gradient of the target magnet was 9.6 [T/m].

\subsubsection{Sextupole Magnet}
The magnetic center of a sextupole magnet in transverse direction ({\it x$_{0}$}, {\it y$_{0}$}) is defined  by the point,
 where the normal quadrupole given by {\it dB$_{x}$/dy}({\it =dB$_{y}$/dx})
 and the skew quadrupole given by {\it dB$_{x}$/dx}({\it =-dB$_{y}$/dy}) are zero.    
Horizontal and vertical distributions of integrated flux densities, {\it B$_{x}$l$_{e}$} and {\it B$_{y}$l$_{e}$} [Tm], in the magnet
 are expressed as,

\begin{equation}\label{bx_sm}
B_{x}(x,y)l_{e}=G_{s}(x-x_{0})(y-y_{0})l_{e}+\delta B _{x,0}l_{e},
\end{equation}
\begin{equation}\label{by_sm}
B_{y}(x,y)l_{e}=\frac{G_{s}}{2}[(x-x_{0})^{2}-(y-y_{0})^{2}]l_{e}+\delta B_{y,0}l_{e},
\end{equation}
where, {\it G$_{s}$} [T/m$^{2}$] is a gradient of a sextupole magnet,
 and {\it $\delta$B$_{x,0}$l$_{e}$} and {\it $\delta$B$_{y,0}$l$_{e}$} [Tm] are dipole components at the magnetic center.
The background field, {\it $\delta$b$_{x}$L} and {\it $\delta$b$_{y}$L}, are also added to Eq.~\ref{bx_sm} and Eq.~\ref{by_sm}. 

A difference of the component {\it B$_{x}$} between two vertical offset positions, {\it y=y’+y$_{f}$} and {\it y’-y$_{f}$},
 is defined as {\it $\Delta$B$_{x,yf}$}.
The difference is shown as,

\begin{equation}\label{bx_diff_sm_x}
\Delta B_{x,yf}(x)l_{e}=2G_{s}y_{f}(x-x_{0})l_{e}.
\end{equation}
Similarly, a difference of the component {\it B$_{x}$} between two horizontal offset positions, {\it x=x’+x$_{f}$} and {\it x’-x$_{f}$},
 is defined as {\it $\Delta$B$_{x,xf}$}.
The difference is shown as,

\begin{equation}\label{bx_diff_sm_y}
\Delta B_{x,xf}(x)l_{e}=2G_{s}x_{f}(y-y_{0})l_{e}.
\end{equation}
A vibration amplitude difference between two vertical offset positions is defined as {\it $\Delta$A$_{y,yf}$},
 and that between two horizontal offset positions is defined as {\it $\Delta$A$_{y,xf}$}.
The amplitude differences, {\it $\Delta$A$_{y,yf}$} and {\it $\Delta$A$_{y,xf}$}, are in proportional to the difference,
 {\it $\Delta$B$_{x,yf}$l$_{e}$} and {\it $\Delta$B$_{x,xf}$l$_{e}$}, respectively.
By scanning the wire in horizontal and vertical directions at above four offset positions, the coordinates
 of the magnetic center were found out.
The offsets, {\it x$_{f}$} and {\it y$_{f}$}, were set to 1 [mm]. 

In the group of multipole magnets, a background field gradient, {\it $\delta$gL} [T], generated by remanent fields
 of other quadrupole magnets was not negligible.
An error field, {\it 2y$_{f}$$\delta$gL} [Tm], is added to Eq.~\ref{bx_diff_sm_x}.
To cancel the background gradient, the {\it Q$_{1}$} magnet excited oppositely as a counter-quadrupole magnet.
Consequently, the amplitude differences, {\it $\Delta$A$_{y,yf}$} and {\it $\Delta$A$_{y,xf}$}, are equal to zero at the magnetic center.
A scan step was 20 [$\mu$m] in the range of $\pm$100 [$\mu$m] in both directions.
Excitation current of all the target magnets was 300 [A].
Nominal gradient of the target magnet was 420 [T/m$^{2}$].

\subsection{Alignment Test}
For all the test magnets, two fiducial points were prepared on the top.
Using these fiducial points, positions of mechanical centers of all the magnets were pre-aligned 
 by a Laser Tracker System (Leica, AT402) and a water level.
A sub-reference line was defined as a line passing through two positions of mechanical centers of the {\it Q$_{1}$}, and {\it Q$_{5}$}.
Then, the mechanical centers of other magnets were adjusted to the sub-reference line by the Tracker.
Yaw angles of these magnets were also corrected by the Tracker.
Pitch and roll angles of these magnets were corrected by the water levels.

Next, positions of the magnetic center of the {\it Q$_{1}$}, and {\it Q$_{5}$} were measured.
A reference line was defined as a line passing through the two positions.
Displacements of the magnetic centers from the reference line were measured for other magnets.
For the {\it Q$_{3}$}, the horizontal position of the magnetic center was adjusted to the reference line
 on a basis of the vibration amplitude.

\subsection{Frequency Tracking and Drift of Magnetic Center}
For both of the signal wire and the feedback wire, the amplitude and the phase were continuously measured in 24 [hours]
 with the basic feedback and without any feedbacks near the fundamental resonance to compare a stability.
The amplitude in both directions were set to 0.1 [mm] by the {\it Sth} and {\it Stv}.
The background gradient was canceled by the counter quadrupole magnet.
A sampling cycle was 3 [sec].
In case of the basic feedback, the feedback cycle was 2.5 [min] using averaged phase by 20 data just before the feedback timing.
A correlation between resonance frequencies of the two wires obtained from the measurement.
Using the correlation, the same continuous measurement was carried out with the advanced feedback.

To observe drifts of the magnetic centers of the {\it Q$_{1}$} and {\it Q$_{3}$}, the vibration amplitudes were continuously measured
 in 50 [hours] just after the excitation with the advaced feedback.
The signal wire was set to the magnetic center, ({\it x},{\it y})=({\it x$_{0}$},{\it y$_{0}$}), at the start of the measurement.
The feedback wire was placed at ({\it x},{\it y})=({\it x$_{0}$}+5[mm],{\it y$_{0}$}-5[mm]).
The current of the feedback wire was 640 [$\mu$A$_{rms}$] for the {\it Q$_{1}$} and 98 [$\mu$A$_{rms}$] for the {\it Q$_{3}$}. 
Changes in the amplitude of the signal wire were converted to changes in the position of the magnetic center.

\section{RESULT}
\subsection{Wire Sag}
Measured wire heights in longitudinal direction are shown in Fig.~\ref{sag_meas_theory}
 with the wire tension of 2.00 [kgw].
Calculated height using measured fundamental resonance frequency is also shown in this figure.
A difference between measured height and calculated one was 1.0 [$\mu$m] at the center of the wire.
The maximum difference was 8.5 [$\mu$m].
In case of the height with the tension of 1.75 [kgw], the difference at the center and the maximum difference were 2.1 [$\mu$m]
 and 9.6 [$\mu$m], respectively.

\begin{figure}[htbp]
\begin{center}
\includegraphics*[width=80mm]{TUO_fukami_sag_meas_theory.eps}
\end{center}
\vspace*{-8mm}
\caption{Wire height distribution in longitudinal direction with the tension of 2.00 [kgw].
Closed circle indicates measured height by the HLS and the WPS. 
Solid line indicates calculated height by measured fundamental resonance frequency.
Open circle and dashed line indicates a difference between the measured height and the calculated one.}
\label{sag_meas_theory}
\end{figure}

\subsection{Magnetic Center and Alignment Test}
As an example for the quadrupole magnet, the vibration amplitudes
 versus the wire positions for the {\it Q$_{3}$} are shown in Fig.~\ref{magnetic_center_qm}.
The wire positions indicate the displacements from the reference line.
The vertical displacement was corrected by the calculated sag.
The displacements in horizontal and vertical directions were (-28.47$\pm$0.07, +53.23$\pm$0.07) [$\mu$m].
After the magnetic center was adjusted to the reference line in horizontal direction,
 the displacements became (-2.59$\pm$0.07, +60.12$\pm$0.04) [$\mu$m]
  (See Fig.~\ref{magnetic_center_qm}, closed circle).

As an example for the sextupole magnet, the amplitude differences
 versus the wire positions for the {\it S$_{2}$} are shown in Fig.~\ref{magnetic_center_sm}.
The displacements were (+28.8$\pm$0.8, +11.8$\pm$0.7) [$\mu$m].
Similarly, for the {\it S$_{4}$}, the displacements were (+36.0$\pm$0.7, +11.1$\pm$0.9) [$\mu$m].

About the measured displacements are not discussed further in this paper because distances between
 the magnetic center and the mechanical center are unknown for the {\it Q$_{1}$} and the {\it Q$_{5}$}.
For all the quadrupole magnets, fitting errors of the magnetic center measurement were less than 0.1 [$\mu$m].
For all the sextupole magnets, the fitting errors were less than 1 [$\mu$m].
A detection limit of an integrated field was estimated to be less than 0.1 [$\mu$Tm].

\begin{figure}[htbp]
\begin{center}
\includegraphics*[width=70mm]{TUO_fukami_q3_bxy_vs_xy.eps}
\end{center}
\vspace*{-8mm}
\caption{Vibration amplitudes versus wire positions for the {\it Q$_{3}$}.
Open and closed circle indicate the amplitudes before and after adjustment by VWM, respectively.
Error bars indicate one standard deviation of ten measurement.
Solid lines indicate results of the least squires fitting.}
\label{magnetic_center_qm}
\end{figure}

\begin{figure}[htbp]
\begin{center}
\includegraphics*[width=70mm]{TUO_fukami_s2_bx_vs_xy.eps}
\end{center}
\vspace*{-8mm}
\caption{Vibration amplitude differences versus wire positions for the {\it S$_{2}$}.
Error bars indicate one standard deviation of ten measurement.
Solid lines indicate results of the least squires fitting.}
\label{magnetic_center_sm}
\end{figure}

\subsection{Temporal changes in the Magnetic Center}
Time dependences on the vibration amplitude and the phase are shown in Fig.~\ref{amp_vs_time} in horizontal direction
 without any feedbacks and with the basic feedback.
The amplitude was decreased by -16 [$\%$] from the original amplitude in 10 [hours] without any feedbacks.
A correlation between the ambient air temperature and the phase was observed clearly
 with a correlation coefficient of 0.89.
Temperature coefficients was 29 [deg/K].

On the other hand, the amplitude and the phase were kept using the basic feedback
 with one standard deviation of 1.6 [$\%$] and 1.7 [deg], respectively.
A correlation between the ambient air temperature and the resonance frequency was observed
 with a correlation coefficient of 0.69.
Temperature coefficients was 0.02 [Hz/K].
The resonance frequency of the signal wire, {\it f$_{s}$}, was strongly correrated with that of the feedback wire, {\it f$_{f}$},
 with a correlation coefficient of 0.99.

From the correlation, for the advanced feedback, a relation between the frequency change of the signal wire, {\it $\Delta$f$_{s}$}
 and that of the feedback wire, {\it $\Delta$f$_{f}$} was determined as {\it $\Delta$f$_{s}$}=0.905{\it $\Delta$f$_{f}$}.
The amplitude and the phase were similarly kept with the advanced feedback using the relation
 with one standard deviation of 1.5 [$\%$] and 1.8 [deg], respectively.
 
Temporal changes in the magnetic center for the {\it Q$_{3}$} is shown in Fig.~\ref{drift_center}
 after starting of excitation.
Temperatures of the yoke and an ambient air are also shown in this figure.
Correlation between the vertical center and the yoke temperature was observed clearly.
However, the yoke temperature was not achieved constant during 2 [days].
The yoke temperature was changed due to a fluctuation of the ambient air temperature.
It is necessary to stabilize the air temperature to estimate a time constant of the drift of the magnetic center.

\begin{figure}[htbp]
\begin{center}
\includegraphics*[width=80mm]{TUO_fukami_amp_phase_x_vs_time.eps}
\end{center}
\vspace*{-8mm}
\caption{Amplitudes and phases of the wire vibrations versus time in horizontal direction without and with feedback.}
\label{amp_vs_time}
\end{figure}

\begin{figure}[htbp]
\begin{center}
\includegraphics*[width=80mm]{TUO_fukami_drift_center.eps}
\end{center}
\vspace*{-8mm}
\caption{Example of temporal change in the magnetic center for the {\it Q$_{3}$}.
Temperature of the yoke and an ambient air are also shown.}
\label{drift_center}
\end{figure}

\section{DISCUSSION}

\subsection{Error budget}
Major error factors are arranged bellow in order making the largest contribution in the alignment error.

\subsubsection{Displacement from theoretical sag}
We are planning that the sag is calculated by measuring only the fundamental resonance in the regular
 alignment procedure without any measurement of the height distrubution wire by wire.
In this case, the difference of an actual wire position from the theoretical curve is contributed directly to the alignment error.
The difference, included any kinks, will be observed from now on.

Since the sag continuously varying along s-axis in a magnet, an effective sag {\it S$_{e}$}
 considered a sensitivity of the vibration should be corrected.
The effective sag is defined as,

\begin{equation}\label{effective_sag}
S_{e}\equiv\frac{1}{l_{e}}\int_{s_u}^{s_d}S(s)sin(\frac{n\pi s}{L})ds,
\end{equation}
Because the effective sag differ from the sag at the center of the magnet, the sag is changed due to the pitch angle.
The change is remarkable for a long magnet, and for a magnet placed near the end of the wire.
For the test magnets, maximum amount of the change is 2.2 [$\mu$m] for the {\it Q$_{5}$}
 with the pitch angle of 0.1 [mrad].    

\subsubsection{Dynamic error caused by drift and deformation}
The magnets should be aligned at their actual excitation current after temperatures of their yoke become stable.
To estimate a time constant of the drift of the magnetic center, an ambient air temperature of the measurement room
 will be stabilized within 0.1 [K].
And then, the error due to the drift is suppressed to less than 1 [$\mu$m] by matching the timing and by shortening
 of the measurement.
In addition, we will observed a change of the magnetic center caused by transportation to the machine tunnel and
 a drift of that caused by deformation after the installation.     

\subsubsection{Systematic error caused by background field}
The background field components, {\it b$_{x}$L} and {\it b$_{y}$L}, were estimated to be -17 and -16 [$\mu$Tm], respectively.
Without cancellation the background, the measured magnetic center of a quadrupole magnet is displaced
 by {\it b$_{y}$L/G$_{q}$l$_{e}$} in horizontal and by {\it b$_{x}$L/G$_{q}$l$_{e}$} in vertical directions.
In case of the {\it Q$_{1}$}, whose effective integrated gradient is minimum of all the test quadrupole magnets,
 these displacements are 12 and 13 [$\mu$m], respectively.
To detect the magnetic center with an accuracy of micro-meter order, it is necessary to reduce the background
 to less than 1/10.

The background field gradient, {\it $\delta$gL}, was estimated to be 42.6 [mT].
The value is equivalent to 0.25 [$\%$] to {\it Q$_{3}$}, which has maximum effective integrated gradient of
 all the test quadrupole magnets.
Without cancellation the background gradient, the measured magnetic center of a sextupole magnet is displaced
 by {\it $\delta$gL/G$_{s}$l$_{e}$} in horizontal direction.
In case of the {\it S$_{2}$}, whose effective integrated gradient is minimum of all the test sextupole magnets,
 the displacement is -480 [$\mu$m].
The displacement is smaller than the value in a strong sextupole magnet
 for the low emittance rings~\cite{ler-spring8-1}-~\cite{ler-sirius}.
However, to detect the magnetic center with an accuracy of micro-meter order, it is necessary to reduce the background
 to less than 1/100.
This reduction is not easy by any magnetic shield.
But, it can be realized by a counter quadrupole magnet using a conventional technique.

\subsubsection{Repeatbilities}
When the tension was released once, and was applied again using same wire,
 the displacements of all the magnets were consistent with the original values within 10 [$\mu$m].  
However, when the wire was removed once, and then, new wire was tensioned again,
 differences in vertical direction exceeded above value.
Maximum difference was 16 [$\mu$m] at the {\it Q$_{3}$}. 
There is a possibility that a sag was changed even though the sag was corrected.
From now on, we are identifying the cause.

\subsubsection{Resolution}
The fitting errors of the magnetic center measurement were much better than aimed accuracy.  
Optimization of the node number and of the applied current, etc., any more improvement is not needed. 

The ambient air temperature of the wire are changed during a scanning period, measured amplitude is devitated.
The temperature coefficient of the phase is strongly depended on a friction at the pulley of the loaded end.
If the pulley was fixed, the coefficient appriximately three times as large as the present value.
Even if the temperature is suddenly changed by 1 [K] without the feedback, an error of the magnetic center
 is estimated to be less than 1 [$\mu$m] for the present coefficient.
In case of the measurement for the magnetic center, the feedback is not required.
 
\subsection{Why B$_{x}$ was chosen for sextupole}
The position of the magnetic center of a sextupole magnet is also determined using {\it B$_{y}$} distributions.
The vibration amplitude {\it A$_{x}$} both in the horizontal and vertical directions follow
 a parabolic function (see Eq.~\ref{by_sm}).
For the {\it S$_{2}$}, the amplitude versus the wire positions are shown in Fig.~\ref{magnetic_center_sm_by}.

The position of the magnetic center was obtained with an error of 2 [$\mu$m].
However, a deviation of the amplitude was much larger than the deviation in the case of {\it B$_{x}$} (Fig.~\ref{magnetic_center_sm}).
It is necessary to widen the scan step and the scan range to suppress the deviation.
In addition, it is unsuitable a {\it in situ} alignment on the basis of the vibration amplitude. 

\begin{figure}[htbp]
\begin{center}
\includegraphics*[width=70mm]{TUO_fukami_s2_by_vs_xy.eps}
\end{center}
\vspace*{-8mm}
\caption{Vibration amplitudes, {\it A$_{x}$}, versus wire positions for the {\it S$_{2}$}.
Error bars indicate one standard deviation of ten measurement.
Solid lines indicate results of the least squires fitting.}
\label{magnetic_center_sm_by}
\end{figure}  

\section{ACKNOWLEDGMENT}
This work was supported by JSPS KAKENHI Grant Number 26390123.




\end{document}
